Author's response to reviews

Title: Effectiveness of management models for facilitating self-management and patient outcomes in adults with diabetes and chronic kidney disease

Authors:

   EDWARD ZIMBUDZI (edward.zimbudzi@monash.edu)
   CLEMENT LO (clement.lo@monash.edu)
   MARIE MISSO (marie.misso@monash.edu)
   SANJEEVA RANASINHA (sanjeeva.ranasinha@monash.edu)
   SOPHIA ZOUNGAS (sophia.zoungas@monash.edu)

Version: 4  Date: 2 June 2015

Author's response to reviews: see over
Dear Editor,

Re: Effectiveness of management models for facilitating self-management and patient outcomes in adults with diabetes and chronic kidney disease

Thank you for the opportunity to re-submit an amended version of our protocol. Below is an itemised response to the reviewer’s comments and resultant amendments.

Two small points addressed as follows:
1. Reference 19 has been changed to PRISMA-P as recommended.
2. The abstract has been amended to indicate the “quality assessment of included studies will be completed”

We believe that the protocol has been enhanced significantly by the reviewer’s comments. Once again, thank you for the opportunity to re-submit the amended version of the protocol for consideration of publication in the Systematic Reviews Journal.

Kind regards,

Corresponding author
Mr. Edward Zimbudzi
Level 1, 43-51 Kanooka Grove
Clayton, VIC 3168, Australia
Australia
T: 0061-3-95943618
F: 0061-3-95946907
Email: edward.zimbudzi@monash.edu